Author’s response to reviews

Title: International students’ experience of a western medical school: A mixed methods study exploring the early years in the context of cultural and social adjustment compared to students from the host country

Authors:

Alice McGarvey (amcgarvey@rcsi.ie)
Elaine Byrne (ebyrne2@rcsi.ie)
Ruairi Brugha (rbrugha@rcsi.ie)
Ronan Conroy (rconroy@rcsi.ie)
Eric Clarke (eclarke@rcsi.ie)

Version: 2 Date: 3 October 2014

Author’s response to reviews:

3rd October 2014

Dear Aldrin

Thank you very much for your prompt feedback. I have made the suggested amendments as follows.

• Highlighted the email addresses for all authors on the title page
• Included author contributions – page 20
• Included acknowledgements – page 20

Yours sincerely

____________________

Alice McGarvey